Author's response to reviews

Title: The role of barrier membranes for guided bone regeneration and restoration of large bone defects: Current experimental and clinical evidence.

Authors:

Rozalia Dimitriou (rozaliadimitriou@yahoo.co.uk)
George I Mataliotakis (george.mataliotakis@gmail.com)
Giorgio M Calori (gmc@studiocalori.it)
Peter V Giannoudis (pgiannoudi@aol.com)

Version: 6 Date: 7 July 2012

Author's response to reviews:

Dear Editors,

Please find submitted the final manuscript. We would like to thank you for your requested formatting changes for our review paper.

Regarding the comment on the references and in particular reference 59 which is cited after reference 142: We confirm that all references are cited in numerical order in the manuscript and tables. Reference 59 is cited after reference 142 in the section of Discussion, after being cited in the text in the section of Types of barriers membranes.... and in Table 1.

Regarding the comment on keywords: We added the keywords after the abstract as requested.

Regarding the comment on the tables: All tables have been added in the manuscript under the corresponding heading. However, these are large tables and may interfere with the flow of the text.

All authors have read and agreed to its content.